Author's response to reviews

Title: Central Obesity and Atherogenic Dyslipidemia in Metabolic Syndrome are Associated with Increased Risk for Colorectal Adenoma in a Chinese Population

Authors:

Chiu-Shong Liu (liucs@ms14.hinet.net)
Chia-Ing Li (a6446@mail.cmuh.org.tw)
Tsai-Chung Li (tcli@mail.cmu.edu.tw)
Wen-Yuan Lin (wylin@mail.cmu.edu.tw)
Tsann Lin (tsann_lin@yahoo.com.tw)
Ya-Chien Chen (n5603@mail.cmuh.org.tw)
Cheng-Chieh Lin (cclin@mail.cmuh.org.tw)

Version: 2 Date: 9 December 2009

Author's response to reviews:

Dear Editor,

We would like to submit a paper “Central Obesity and Atherogenic Dyslipidemia in Metabolic Syndrome are Associated with Increased Risk for Colorectal Adenoma in a Chinese Population”, which was enclosed, to the BMC GASTROENTEROLOGY as an original article.

Recent epidemiological studies have reported a positive correlation between metabolic syndrome and the presence of colorectal adenoma. In order to prevent colorectal cancer, it is crucial to identify the risk factors that are associated with the development of colorectal neoplasia. Therefore, we examined the relationship between the components of metabolic syndrome and prevalence of colorectal adenoma in a Chinese population. The authors declare that the manuscript has not been published previously, and is not currently submitted for publication elsewhere.

Your sincerely

Dr. Cheng-Chieh Lin

Dept. of Family Medicine

China Medical University Hospital